Author’s response to reviews

Title: Family physician views about primary care reform in Ontario: a postal questionnaire

Authors:

Dr Duncan J W Hunter (hunter@post.queensu.ca)
Samuel E D Shortt (seds@post.queensu.ca)
Peter M Walker (petewalker007@yahoo.ca)
Dr Marshall Godwin (godwinm@post.queensu.ca)

Version: 6 Date: 26 Feb 2004

Minor editing changes to the title page:

clarification of affiliations

inclusion of "*" to denote contributions.